Author's response to reviews

Title: Relationship of aerobic fitness and motor skills with memory and attention in preschoolers (Ballabeina): A cross-sectional and longitudinal study

Authors:

Iris Niederer (iris.niederer@unibas.ch)
Susi Kriemler (susi.kriemler@unibas.ch)
Janine Gut (janine.gut@unibas.ch)
Tim Hartmann (tim.hartmann@unibas.ch)
Christian Schindler (christian.schindler@unibas.ch)
Jérôme Barral (jerome.barral@unil.ch)
Jardena J Puder (jardena.puder@chuv.ch)

Version: 4 Date: 20 April 2011

Author's response to reviews:

Dear Editorial Production Team

Thank you for your feedback. I've done the following corrections:
- I've now added the sentence in the authors' contributions.
- I've removed all highlighting and tracking from the manuscript.
- I've corrected the reference list and checked the manuscript for typographical errors.
- I've corrected HV-VK into KHV-VK.

Kind regards, Iris Niederer